Author’s response to reviews

Title: Case report: Ocular toxicity due to Trametinib and Dabrafenib

Authors:

Stephanie Sarny (stephanie@sarny.at)
Michael Neumayer (michael.neumayer@kabeg.at)
Julian Kofler (julian.kofler@kabeg.at)
Yosuf El-Shabrawi (yosuf.elshabrawi@kabeg.at)

Version: 1 Date: 20 Apr 2017

Author’s response to reviews:

Dear Editor,

may I kindly ask you to consider the manuscript “Case report: Ocular toxicity due to Trametinib and Dabrafenib” for publication in the BMC Ophthalmology.

We report a case of a patient suffering from severe ophthalmologic complications under therapy with Trametinib and Dabrafenib. After several months, the patient suffered from serous neuroretinal detachment and uveitis anterior. The therapy was stopped immediately, however, retinal complications appeared again as the therapy was restarted after a year. A habituation effect was not observed.

We would be very happy to have the accept of BMC Ophthalmology to published this manuscript.

Thank you very much for your consideration.

Yours sincerely,

Stephanie Sarny, MD PhD